Author’s response to reviews

**Title:** Multiple constraints compromise decision-making about implantable medical devices for individual patients: qualitative interviews with physicians

**Authors:**

Anna Gagliardi (anna.gagliardi@uhnresearch.ca)

Ariel Ducey (aducey@ucalgary.ca)

Pascale Lehoux (pascale.lehoux@umontreal.ca)

Thomas Turgeon (tturgeon@orthoinno.com)

Jeremy Kolbunik (jeremy.kolbunik@sinaihealthsystem.ca)

Sue Ross (sue.ross@albertahealthservices.ca)

Patricia Trbovich (patricia.trbovich@utoronto.ca)

Anthony Easty (tony.easty@utoronto.ca)

Chaim Bell (chaim.bell@sinaihealthsystem.ca)

David Urbach (david.urbach@uhn.ca)

**Version:** 4 **Date:** 26 Nov 2017

**Author’s response to reviews:**

EDITORIAL REQUEST

Please include in the 'Ethics approval and consent to participate' section the name of the University to which 'University Health Network Research Ethics Board' is associated.

AUTHOR RESPONSE

The ethics statement in the Declarations section (page 18) was amended to: "The University Health Network (UHN) Research Ethics Board approved this study. UHN is a multi-hospital corporation affiliated with the University of Toronto but undertakes its own ethics review. Participants provided written informed consent prior to being interviewed."